

Distributed Global Debris Thickness Estimates Reveal Debris Significantly Impacts Glacier Mass Balance

D. Rounce, R. Hock, R. Mcnabb, R. Millan, C. Sommer, M. Braun, P. Malz,

F. Maussion, J. Mouginot, T. Seehaus, et al.

▶ To cite this version:

D. Rounce, R. Hock, R. Mcnabb, R. Millan, C. Sommer, et al.. Distributed Global Debris Thickness Estimates Reveal Debris Significantly Impacts Glacier Mass Balance. Geophysical Research Letters, 2021, 48 (8), 10.1029/2020GL091311. hal-03403505

HAL Id: hal-03403505 https://hal.science/hal-03403505v1

Submitted on 30 Oct 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - ShareAlike 4.0 International License

<mark>-</mark>



Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL091311

Key Points:

- We produce the first distributed global debris thickness estimates
- Accounting for debris significantly reduces regional glacier mass loss
- The similar thinning rates of debris-covered and clean ice glaciers in High Mountain Asia is primarily caused by differences in ice dynamics

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. R. Rounce, drounce@cmu.edu

Citation:

Rounce, D. R., Hock, R., McNabb, R. W., Millan, R., Sommer, C., Braun, M. H., et al. (2021). Distributed global debris thickness estimates reveal debris significantly impacts glacier mass balance. *Geophysical Research Letters*, 48, e2020GL091311. https://doi. org/10.1029/2020GL091311

Accepted 23 FEB 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Distributed Global Debris Thickness Estimates Reveal Debris Significantly Impacts Glacier Mass Balance

D. R. Rounce^{1,2}, R. Hock^{2,3}, R. W. McNabb^{3,4}, R. Millan⁵, C. Sommer⁶, M. H. Braun⁶, P. Malz⁶, F. Maussion⁷, J. Mouginot^{5,8}, T. C. Seehaus⁶, and D. E. Shean⁹

¹Department of Civil and Environmental Engineering, Carnegie Mellon University, Pittsburgh, PA, USA, ²Geophysical Institute, University of Alaska Fairbanks, Fairbanks, AK, USA, ³Department of Geosciences, University of Oslo, Oslo, Norway, ⁴School of Geography and Environmental Sciences, Ulster University, Coleraine, UK, ⁵Université Grenoble Alpes, CNRS, IRD, Grenoble INP, IGE, Grenoble, France, ⁶Institut für Geographie, Friedrich-Alexander-Universität Erlangen-Nürnberg, Erlangen, Germany, ⁷Department of Atmospheric and Cryospheric Sciences, University of Innsbruck, Innsbruck, Austria, ⁸Department of Earth System Science, University of California, Irvine, CA, USA, ⁹Department of Civil and Environmental Engineering, University of Washington, Seattle, WA, USA

Abstract Supraglacial debris affects glacier mass balance as a thin layer enhances surface melting, while a thick layer reduces it. While many glaciers are debris-covered, global glacier models do not account for debris because its thickness is unknown. We provide the first globally distributed debris thickness estimates using a novel approach combining sub-debris melt and surface temperature inversion methods. Results are evaluated against observations from 22 glaciers. We find the median global debris thickness is ~0.15 \pm 0.06 m. In all regions, the net effect of accounting for debris is a reduction in sub-debris melt, on average, by 37%, which can impact regional mass balance by up to 0.40 m water equivalent (w.e.) yr⁻¹. We also find recent observations of similar thinning rates over debris-covered and clean ice glacier tongues is primarily due to differences in ice dynamics. Our results demonstrate the importance of accounting for debris in glacier modeling efforts.

Plain Language Summary Many glaciers around the world have a layer of debris (boulders, rocks, and sand) covering the underlying ice over much of the glacier surface, yet global glacier models do not account for debris because the debris thickness is unknown. Here we provide the first estimates of debris thickness for debris-covered glaciers globally and show the debris substantially reduces regional glacier mass loss. We also find that recent observations that debris-covered and clean ice glaciers are thinning at similar speeds is primarily due to differences in how glaciers flow. Our results fundamentally advance our ability to account for debris in glacier reconstructions, landscape evolution models, hazard assessments, and glacier projections of glacier runoff and their contribution to sea-level rise.

1. Introduction

The thickness of supraglacial debris will control how glaciers respond to climate change, since a thin layer (less than a few centimeters) will enhance melting, while a thick layer will insulate the underlying ice and suppress melting (Østrem, 1959). While this effect is well known at the point scale, recent studies in High Mountain Asia found similar elevation change rates for debris-covered and clean-ice glaciers (Gardelle et al., 2013; Kääb et al., 2012), a phenomenon referred to as the "debris-cover anomaly" (Pellicciotti et al., 2015). Proposed explanations include debris-covered glaciers experiencing enhanced melting due to ice cliffs and ponds and/or having comparatively lower emergence velocities (Brun et al., 2018). A regional study that accounted for debris on High Mountain Asia glaciers found debris delays projected mass loss (Kraaijenbrink et al., 2017). However, the effects of debris on regional and global projections of glacier mass change are largely unknown due to a lack of accurate debris thickness estimates. Given debris covers ~4%-7% of the global glacier area (excluding the ice sheets), and up to 23% regionally (Herreid & Pellicciotti, 2020; Scherler et al., 2018), accurate debris thickness estimates are urgently needed to quantify the impact of debris on glacier mass balance from local to global scales.

Despite the importance of debris thickness for glacier evolution, in-situ measurements of debris thickness and sub-debris melt rates are relatively scarce and primarily limited to glaciers in Central Europe and High Mountain Asia (Tables S1 and S2). Even in-situ measurements from well-studied debris-covered glaciers are

typically spatially and/or temporally limited. Nonetheless, these measurements have been used to develop empirical relationships to estimate debris thickness (Kraaijenbrink et al., 2017; Mihalcea et al., 2008), or to validate debris thickness estimates from physically-based surface temperature inversion (Foster et al., 2012; Rounce & McKinney, 2014) or sub-debris melt inversion (Rounce et al., 2018) methods. At present, the sub-debris melt inversion method is the only physically-based method that can accurately estimate debris thicknesses greater than 0.5 m. However, the method is currently limited to elevation bins and requires accurate estimates of the climatic mass balance (i.e., the sum of surface accumulation, ablation, and refreezing), which must be derived from limited in-situ measurements or remotely-sensed elevation change data. The latter must be corrected for elevation change due to ice flow, which is challenging over steep, fast-flowing areas. Conversely, surface temperature inversion methods have no spatial limitations but struggle to estimate thick debris and must accurately resolve the surface energy balance fluxes at the time when the satellite image is acquired.

Here we produce the first global debris thickness estimates for each of the 92,033 debris-covered glaciers in the Randolph Glacier Inventory (RGI, version 6) (RGI Consortium, 2017; Scherler et al., 2018), excluding the ice sheets and Antarctic periphery. These estimates are then used to determine the cause of the debris cover anomaly and quantify how debris impacts sub-debris melt and glacier mass balance on local, regional, and global scales.

2. Methods

Our novel approach first uses a sub-debris melt inversion method to estimate the debris thickness over binned near-stagnant debris-covered areas for all debris-covered glaciers greater than 2 km^2 with available geodetic mass balance and surface velocity data (Figure S1). A surface temperature inversion method is then used to estimate the distributed debris thickness over the glacier's entire debris-covered area using the binned debris thickness estimates from the sub-debris melt inversion method for calibration. For the remaining debris-covered glaciers (47% of the total debris-covered area), we extrapolate using the nearest calibrated glaciers' coefficients to estimate their distributed debris thickness.

2.1. Sub-debris Melt Inversion Method

The sub-debris melt inversion method relates annual sub-debris melt (M; m w.e.) to debris thickness (h_d ; m) according to a second-order reaction rate equation:

$$M = \frac{M_0}{\left(1 + kM_0h_d\right)} \tag{1}$$

where M_0 and k are coefficients derived from fitting the modeled sub-debris melt from a debris-covered glacier energy balance model (Rounce et al., 2015; Text S1) for various debris thicknesses (Figure S2a).

Since debris is located in the ablation area, we can assume the sub-debris melt is equivalent to the annual climatic mass balance, which is estimated as the elevation change plus the flux divergence (all in m w.e.). We refer to this as the observed sub-debris melt. Elevation change data between 2000 and 2018 (Braun et al., 2019; Malz et al., 2019; Seehaus et al., 2019; Shean et al., 2020; Sommer et al., 2020) was supplemented with data from Advanced Spaceborne Thermal Emission and Reflection Radiometer (ASTER) DEMs, Arctic DEM v3.0 strips, and SPOT5 IPY-SPIRIT DEMs (Text S2; Table S3). Following Rounce et al. (2018), the flux divergence is estimated from surface velocity data from 1985 to 2018 (Gardner et al., 2019) and ice thickness estimates (Farinotti et al., 2019), assuming only internal deformation and thus a ratio of column-average ice velocity to surface velocity of 0.8. Missing velocity data are supplemented by feature tracking of Landsat and Sentinel-2 images between 2015 and 2018 (Millan et al., 2019) (Table S3).

The inversion is only performed over near-stagnant ($<7.5 \text{ m yr}^{-1}$) debris-covered areas, and the observed sub-debris melt data are aggregated into 10 m elevation bins, to minimize uncertainty associated with the flux divergence (since the flux divergence is near zero). Any bins where the observed sub-debris melt exceeds our maximum modeled sub-debris melt, or is below our minimum modeled sub-debris melt, are discarded. The latter helps avoid estimating the debris thickness over relatively stable non-glacierized terrain

due to glacier retreat, since the debris cover and glacier extents were delineated over different time periods (RGI Consortium, 2017; Scherler et al., 2018). Uncertainty arising from the elevation change, surface velocity, and ice thickness data is incorporated into the debris thickness estimates (Text S3).

2.2. Surface Temperature Inversion Method

The distributed debris thickness is then estimated for each debris-covered glacier grid cell using the surface temperature inversion method, based on modeled surface temperatures and Landsat-8 surface temperature data, that is calibrated with the debris thickness estimates from the sub-debris melt inversion method. The surface temperature data is a composite image generated in Google Earth Engine from all available Landsat-8 images between 2013 and 2018 from the ablation season (Kraaijenbrink et al., 2017). The relationship between the debris thickness and modeled debris surface temperature (T_s ; °C) from a debris-covered glacier energy balance model (Rounce et al., 2015; Text S1) (Figure S2c) is estimated using the Hill equation:

$$h_d = \left(\frac{T_s b^c}{a - T_s}\right)^{\frac{1}{c}} \tag{2}$$

where a, b, and c are calibrated coefficients that capture the nonlinear relationship between the debris thickness and surface temperature at the time when the data were acquired. To avoid overestimating the debris thickness by only calibrating the coefficients with near-stagnant bins at the terminus that likely comprise thicker debris, we assume the coldest debris-covered elevation bin has a debris thickness of 0.02 m. Furthermore, the calibration minimizes the difference between the estimated debris thicknesses' sub-debris melt rates from the two inversion methods (Equation 1). This gives more weight to thinner debris, which is appropriate since the sub-debris melt inversion method is more sensitive to thinner debris.

To capture the nonlinear relationship between the debris thickness and surface temperature, we identify all snow-free days, according to the energy balance model, within one month of the Landsat-8 acquisition date. Since the debris is warming rapidly in the morning when the satellite image is acquired (~10:30 a.m. local), we linearly interpolate the hourly modeled surface temperatures to estimate the surface temperatures that coincide with the acquisition time. We then fit coefficients *a*, *b*, and *c* using the interpolated modeled surface temperatures; thus, producing up to 60 potential sets of coefficients. We estimate the debris thickness from the Landsat-8 surface temperature data for each set and select the coefficients that minimize the difference between the estimated sub-debris melt rates and those from the sub-debris melt inversion method (Equation 1). To ensure the difference between the sub-debris melt and surface temperature inversion methods is minimized, we then fix coefficients *b* and *c*, and recalibrate *a*. The calibrated coefficients are then used to estimate the distributed debris thickness over the entire debris-covered glacier area (Figure S1).

2.3. Extrapolated Debris Thickness Estimates

For glaciers lacking calibration data (47% of the total debris-covered area), that is, those that are missing data, too small, or not stagnant at their terminus, we use coefficients (M_0 , k, a, b, and c) from the nearest calibrated glaciers to estimate the debris thickness. To ensure the extrapolated debris thicknesses are physically reasonable, each uncalibrated glacier loops through the parameter sets of its nearest neighbors and accepts the debris thickness estimates if they meet two conditions: (i) the debris thickness estimate for the coldest elevation bin, that is, the thinnest debris, is less than 0.2 m, and (ii) the debris thickness estimate for the warmest elevation bin, that is, the thickest debris, is less than 3.0 m. The uncalibrated glaciers loop through the parameter sets until 10 reasonable estimates are found. The debris thickness is calculated to be the median of these 10 reasonable estimates for each grid cell. A leave-one-glacier-out cross-validation, that is, applying the extrapolation to calibrated glaciers, reveals the median (\pm normalized median absolute deviation) difference is 0.04 ± 0.20 m. The median distance to the nearest calibrated glacier is 12 km (Figure S3).

2.4. Sub-debris Melt Enhancement Factors

Sub-debris melt enhancement factors are estimated from the debris thickness using the debris thickness-melt curves (Equation 1) normalized by the clean-ice melt (Figure S2b). For debris thicknesses less than 0.02 m, the enhancement factor is linearly interpolated between the value of one for clean ice and the enhancement factor associated with a debris thickness of 0.02 m. For each climate grid cell containing debris-covered glaciers, clean-ice melt is estimated using an energy balance model (Text S1).

2.5. Modeled Regional Mass Balance

The Python Glacier Evolution Model (PyGEM) (Rounce et al., 2020) is used to quantify the impact of debris thickness on regional mass balance for Central Europe, Caucasus/Middle East, and High Mountain Asia (Central Asia, South Asia West, South Asia East). PyGEM estimates the climatic mass balance of every glacier using 10 m elevation bins and a monthly time step. The model is forced with monthly ERA5 air temperature and precipitation data, and computes glacier melt using a degree-day model, accumulation using a temperature threshold, and refreezing based on the annual air temperature. The sub-debris melt enhancement factors for each elevation bin are used to account for debris. The glacier area is assumed constant.

We calibrate each glacier using the calibration scheme from Huss and Hock (2015) and regional specific mass balance data from 2000 to 2018 for High Mountain Asia (Shean et al., 2020) and 2006–2015 for the other regions (Hock et al., 2019). We then run the calibrated model from 2000 to 2018 without applying the enhancement factors for debris-covered elevation bins and compare the results to quantify the impact of accounting for debris. We perform an additional simulation in High Mountain Asia, where we calibrate the model using data from every glacier to determine how the use of regional mass balance data for calibration impacts the results. The regional specific mass balances are within ± 0.01 m w.e. yr⁻¹ compared to the simulations run using regional specific mass balance for calibration, thus justifying the use of regional data for the other regions. Uncertainties associated with the debris thickness estimates are evaluated from additional simulations using enhancement factors derived from the 16% and 84% debris thickness estimates (Text S3).

3. Results

3.1. Debris Thickness Estimates

The distributed debris thickness estimates of some of the most well-studied debris-covered glaciers in Alaska, Europe, and High Mountain Asia show that debris thickness is less than 0.02 m in the uppermost debris-covered areas and increases toward the termini of the glaciers (Figure 1) as expected (Anderson & Anderson, 2018). The maximum debris thickness of these glaciers ranges from \sim 0.6 m on Kennicott and Lirung to more than 2 m on Koxkar and Ngozumpa. The debris thickness of most glaciers also increases toward the lateral margins, where unstable valley walls can provide an additional source of debris. Glaciers with prominent medial moraines may have thicker debris in the center of the glacier (e.g., Suldenferner; Figure 1b).

To evaluate our debris thickness estimates, we compare individual measurements with geolocation information to the nearest pixel and also aggregate measurements/pixels into 50 m elevation bins comparing those with more than five observations (Table S1; Figures 2a and 2b). Both comparisons face issues with measurements being limited in space and/or quantity, varying considerably over small scales (<30 m), having limited geolocation information, and changing over time due to glacier flow and/or local debris redistribution. Our estimates agree reasonably well with these in-situ measurements (Figures 2a and 2b; Table S4) considering the method's global application. The scatter is expected considering the small-scale variability with point measurements and that its unlikely the limited in-situ measurements are representative of the entire elevation bin which includes thicker debris near the lateral moraines (Text S4; Figure S4). Interestingly, our debris thickness uncertainty estimates (Text S3; Figure S5) are comparable to the root mean square error of in-situ versus modeled debris thicknesses, regardless of debris thickness magnitude (Table S4), which lends confidence to our uncertainty estimates.





Figure 1. Modeled debris thickness (h_d) for well-studied debris-covered glaciers located in Alaska (a), Central Europe (b, c), Central Asia (d), South Asia West (f), and South Asia East (e, g, h). Note the saturated, nonlinear colorbar. Accumulation areas, where debris cover is absent, are not shown. Arrows indicate glacier flow direction toward the terminus.

The method's ability to reproduce debris thicknesses exceeding 1 m (e.g., Koxkar (Wu & Liu, 2012), Ngozumpa (Nicholson & Benn, 2012); Figure 1) suggests the method can reasonably estimate the total volume of debris. Similarly, despite a nonlinear debris thickness-melt relationship, the method's ability to reproduce debris thicknesses less than 0.2 m (e.g., Kennicott in Alaska (Anderson et al., 2019), Suldenferner in Central



Figure 2. Comparison of modeled and observed (a, b) debris thickness (h_d) for elevation bins and individual points/pixels and (c) sub-debris melt rates using the observed debris thickness. Binned debris thicknesses are scaled and colored by the number of observations (n_{obs}). Sub-debris melt rates are colored by debris thickness. Error bars represent spatial variability within the corresponding 50 m elevation bin according to the median and normalized median absolute deviation. Details of the observations are provided in Tables S1 and S2.



Geophysical Research Letters



Figure 3. Regional debris thickness (h_d) and sub-debris melt enhancement factors (E_d). Bars show median values with the tick showing values of 0.1 m and 0.5, respectively. Circles show relative debris-covered area (Scherler et al., 2018). Histograms are normalized by their total area. Enhancement factors greater than one are darker red. Numbers refer to the total debris-covered area (km²). Debris-covered glaciers are shown in blue and RGI regions are outlined in black.

Europe (Nicholson & Boxall, 2020), Khumbu in High Mountain Asia (Rowan & Gibson, 2020); Figure 1) suggests sub-debris melt rates are well estimated.

The global debris thickness estimates show considerable regional-scale variability (Figure 3). The global median (16%-84% confidence interval) debris thickness of all debris-covered areas is 0.15 m (0.09–0.21 m) and the total volume of debris is 7.3 km³ (4.3–12.8 km³). The global mean debris thickness of 0.35 m (0.21–0.62 m) is higher than the median since the debris thickness histograms are right-skewed (Figure 3). Regionally, the median debris thickness varies from 0.04 m (0.02–0.07 m) in North Asia to 0.19 m (0.12–0.27 m) in South Asia East. Other regions with thicker debris include Alaska and Caucasus/Middle East. Regions with thinner debris include Svalbard and Southern Andes.

3.2. Debris Thickness Impact on Sub-debris Melt

The impact of debris thickness on sub-debris melt is quantified using sub-debris melt enhancement factors (hereafter referred to as enhancement factors), which equal the ratio between sub-debris melt and clean-ice melt. Enhanced sub-debris melt beneath thin debris (typically 0.03–0.05 m) coincides with enhancement factors greater than 1, while thick debris can have enhancement factors less than 0.1, corresponding to a greater than 90% reduction in sub-debris melt. The agreement and lack of systematic bias between modeled sub-debris melt rates and in-situ measurements (Figure 2; r = 0.85, RMSE = 0.08 mm w.e. d⁻¹) provides confidence in our enhancement factors. Note the model underestimates high melt rates (>50 mm w.e. d⁻¹), which may be due to local topography or differences in debris properties.

For individual glaciers, spatial variations in modeled debris thickness results in enhanced sub-debris melting in the uppermost debris-covered areas and limited melting near the glacier termini (Figure S6). Some debris-covered glaciers (e.g., Koxkar, Ngozumpa) have enhancement factors less than 0.5 over up to 84% of the debris-covered area. Others (e.g., Kennicott, Miage, Baltoro) have values greater than one over up to 45% of the debris-covered area.

Regionally, the nonlinear debris thickness-melt relationship means that the spatial distribution of debris thickness, and corresponding enhancement factors, are important (Figure 3). The global mean enhancement factor is 0.63 (0.51–0.73), varying from a minimum of 0.39 (0.28–0.5) in Iceland to a maximum of 0.80 (0.68–0.88) in North Asia. In all regions, the net effect of accounting for debris is a reduction in sub-debris glacier melt.

3.3. Debris Thickness Impact on Regional Mass Balance

The impact of debris on regional glacier mass balance is exemplified for some regions with significant debris cover: Central Europe, Caucasus/Middle East, Central Asia, South Asia West, and South Asia East. The net impact depends on the spatial distribution of debris thickness, relative amount of debris-covered area to total glacier area, and mass balance gradient. The latter is essential since the glacier termini, where debris is thickest, coincides with where the climatic mass balance is most negative for clean-ice glaciers. We estimate how much accounting for debris thickness affects regional specific mass balance (i.e., mass change per unit area) by running simulations that (1) account for debris thickness using enhancement factors and (2) assume no debris cover.

Accounting for debris changes the regional specific mass balance by $+0.40 \text{ m w.e. yr}^{-1}$ (0.25–0.64 m w.e. yr⁻¹) for Caucasus/Middle East and $+0.18 \text{ m w.e. yr}^{-1}$ (0.06–0.38 m w.e. yr⁻¹) for Central Europe (Figure 4). The change in regions in High Mountain Asia varies from $+0.26 \text{ m w.e. yr}^{-1}$ (0.19–0.36 m w.e. yr⁻¹) in South Asia East to $+0.07 \text{ m w.e. yr}^{-1}$ (0.03–0.12 m w.e. yr⁻¹) in South Asia West; interestingly, the changes as a percentage of regional mass balance are similar, varying from +30 to 39%. These results show that debris substantially reduces regional mass loss despite covering a small fraction of the total glacier area (5%–23%).

3.4. Debris Cover Anomaly

Our debris thicknesses are derived from elevation change and surface temperature data that include surface features like ice cliffs and ponds, which can melt 3–17 times faster than the surrounding debris-covered ice (Brun et al., 2018; Miles et al., 2018). Consequently, the mixed-pixel effects reduce modeled debris thickness estimates (Rounce et al., 2018) and increase enhancement factors (Text S4). Despite their higher melt rates, the enhancement factors for cliffs and ponds on debris-covered glaciers in previous studies vary from 0.81 to 1.85 (Table S5), which are comparable to our enhancement factors for thin debris (Figure S6).

In High Mountain Asia, the mean enhancement factor of 0.64 indicates melt beneath debris-covered areas is, on average, reduced by 36% compared to clean ice. If we assume ice cliffs and ponds have enhancement factors of 0.81–1.85 and comprise 6%–14% of the debris-covered areas (Brun et al., 2018; Herreid & Pellicciotti, 2018; Miles et al., 2018), then enhancement factors for the remaining debris-covered area are 0.44–0.63. We can therefore estimate that enhanced melting due to cliffs and ponds accounts for 3%–35% (0.01–0.20 of the 0.37–0.56 reduction) of the debris cover anomaly and attribute the remaining 65%–97% to reduced emergence velocities over debris-covered areas compared to clean ice. Hence, less ice flux into the





Figure 4. Climatic mass balance (b_{clim}), area-altitude distributions (*AAD*), and sub-debris melt enhancement factors (E_d) as a function of elevation for various regions. Climatic mass balances from PyGEM are shown accounting for debris (green) and assuming clean ice (blue) with colored text showing the regional glacier-wide mass balance (m w.e. yr⁻¹). Area-altitude distributions are shown for the entire glacier (blue) and debris-covered area (AAD_d ; green). Shaded areas represent uncertainty (16% and 84% confidence interval) from debris thickness estimates. Dotted horizontal line shows the modeled mean annual equilibrium line altitude.

slow-moving, debris-covered portion of the glacier is the dominant cause of the debris cover anomaly in High Mountain Asia.

4. Conclusions

Our global debris thickness estimates provide the first assessment of how debris affects the sub-debris melt of debris-covered glaciers at local, regional, and global scales. At the glacier scale, the spatial variations in debris thickness can cause glacier tongues to have an inverted ablation gradient meaning the glacier may experience higher rates of mass loss further upglacier than at its terminus, which is consistent with previous studies (e.g., Benn & Lehmkuhl, 2000). This can cause glaciers to stagnate, develop supraglacial and proglacial lakes, and become a flood hazard for downstream communities (Benn et al., 2012; Rounce et al., 2017; Shugar et al., 2020). These proglacial lakes also substantially increase glacier-wide mass loss

rates (King et al., 2019). Our enhancement factors will therefore enable glacier evolution models to account for these inverted ablation gradients and help predict the development of glacial lakes and their impact on glacier-wide mass change.

At the regional scale, enhancement factors reveal the net effect of accounting for debris is a reduction in sub-debris melt in all regions, which can significantly impact regional mass balance. Furthermore, we find the primary cause of the debris cover anomaly is a reduction in ice flux into the lowermost portions of debris-covered glaciers. Regional glacier models should therefore seek to more accurately account for glacier dynamics. These enhancement factors will also improve ice thickness estimates for debris-covered glaciers, which currently do not account for debris (Farinotti et al., 2017). Future work should prioritize systematically measuring debris thickness to enable proper validation and error assessments and measuring sub-debris melt beneath thin debris to determine local and regional variability of the critical debris thickness (e.g., Fyffe et al., 2020). Furthermore, advancing models' ability to estimate melt beneath thin debris, improving the meteorological forcing, incorporating site-specific measurements of debris properties and their spatial variation, and developing ways to for account for local topography may significantly improve and reduce uncertainty associated with the debris thicknesses and enhancement factors.

Overall, these global debris thickness estimates fundamentally advance our ability to account for debris thickness in glacier reconstructions, debris transport models, hazard assessments, and landscape evolution models. Future work should seek to account for debris in global glacier evolution models to determine how debris cover will affect projections of glacier mass change and runoff at regional and global scales.

Data Availability Statement

The datasets generated for this study can be found in the National Snow and Ice Data Center (NSIDC) (https://nsidc.org/data/HMA_DTE/). The model code is available from a permanent DOI repository (https://zenodo.org/record/4644355#.YGHozV1Kj0o). The glacier masks and DEM maps were generated with OGGM (Maussion et al., 2019, v1.3.2 available at https://zenodo.org/record/4588404#.YF8-fxIo9hF).

References

Anderson, L. S., & Anderson, R. S. (2018). Debris thickness patterns on debris-covered glaciers. *Geomorphology*, 311, 1–12. https://doi. org/10.1016/j.geomorph.2018.03.014

- Anderson, L. S., Anderson, R. S., Buri, P., & Armstrong, W. H. (2019). Debris cover and the thinning of Kennicott Glacier, Alaska, Part A: In situ mass balance measurements. *The Cryosphere Discussions*, 1–28. https://doi.org/10.5194/tc-2019-174
- Benn, D. I., Bolch, T., Hands, K., Gulley, J., Luckman, A., Nicholson, L. I., et al. (2012). Response of debris-covered glaciers in the Mount Everest region to recent warming, and implications for outburst flood hazards. *Earth-Science Reviews*, 114(1–2), 156–174. https://doi. org/10.1016/j.earscirev.2012.03.008
- Benn, D. I., & Lehmkuhl, F. (2000). Mass balance and equilibrium-line altitudes of glaciers in high-mountain environments. *Quaternary* International, 65–66, 15–29. https://doi.org/10.1016/s1040-6182(99)00034-8
- Braun, M. H., Malz, P., Sommer, C., Farías-Barahona, D., Sauter, T., Casassa, G., et al. (2019). Constraining glacier elevation and mass changes in South America. *Nature Climate Change*, 9(2), 130–136. https://doi.org/10.1038/s41558-018-0375-7
- Brun, F., Wagnon, P., Berthier, E., Shea, J. M., Immerzeel, W. W., Kraaijenbrink, P. D. A., et al. (2018). Ice cliff contribution to the tongue-wide ablation of Changri Nup Glacier, Nepal, central Himalaya. *The Cryosphere*, 12(11), 3439–3457. https://doi.org/10.5194/ tc-12-3439-2018
- Farinotti, D., Brinkerhoff, D. J., Clarke, G. K. C., Fürst, J. J., Frey, H., Gantayat, P., et al. (2017). How accurate are estimates of glacier ice thickness? Results from ITMIX, the Ice Thickness Models Intercomparison eXperiment. *The Cryosphere*, 11(2), 949–970. https://doi. org/10.5194/tc-11-949-2017
- Farinotti, D., Huss, M., Fürst, J. J., Landmann, J., Machguth, H., Maussion, F., & Pandit, A. (2019). A consensus estimate for the ice thickness distribution of all glaciers on Earth. *Nature Geoscience*, *12*(3), 168–173. https://doi.org/10.1038/s41561-019-0300-3
- Foster, L. A., Brock, B. W., Cutler, M. E. J., & Diotri, F. (2012). A physically based method for estimating supraglacial debris thickness from thermal band remote-sensing data. *Journal of Glaciology*, 58(210), 677–691. https://doi.org/10.3189/2012JoG11J194
- Fyffe, C. L., Woodget, A. S., Kirkbridge, M. P., Deline, P., Westoby, M. J., & Brock, B. W. (2020). Processes at the margines of supraglacial debris cover: Quantifying dirty ice ablation and debris redistribution. *Earth Surface Processes and Landforms*, 45, 2272–2290. https:// doi.org/10.1002/esp.4879
- Gardelle, J., Berthier, E., Arnaud, Y., & Kääb, A. (2013). Region-wide glacier mass balances over the Pamir–Karakoram–Himalaya during 1999–2011. *The Cryosphere*, 7(4), 1263–1286. https://doi.org/10.5194/tc-7-1263-2013
- Gardner, A. S., Fahnestock, M. A., & Scambos, T. A. (2019). ITS_LIVE regional glacier and ice sheet surface velocities. Data archived at National Snow and Ice Data Center. https://doi.org/10.5067/6II6VW8LLWJ7
- Herreid, S., & Pellicciotti, F. (2018). Automated detection of ice cliffs within supraglacial debris cover. *The Cryosphere*, *12*(5), 1811–1829. https://doi.org/10.5194/tc-12-1811-2018
- Herreid, S., & Pellicciotti, F. (2020). The state of rock debris covering Earth's glaciers. *Nature Geoscience*, 13(9), 621–627. https://doi.org/10.1038/s41561-020-0615-0

Acknowledgments

We thank Mohan Chand, Astrid Lambrecht, Christoph Mayer, and Yong Zhang for providing ablation data and Leif Anderson, Lesley Foster, Catriona Fyffe, Wilfried Hagg, Peter Moore, and Lindsey Nicholson for providing debris thickness data. High-performance computing and data storage resources were provided by the Research Computing Systems Group at the University of Alaska Fairbanks. Supplemental surface velocities were produced using the GRICAD infrastructure, which is supported by Grenoble research communities.



- Hock, R., Rasul, G., Adler, C., Caceres, B., Gruber, S., Hirabayashi, Y., et al. (2019). High mountain areas. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC special report on the ocean and cryosphere in a changing climate.* Sine loco: Intergovernmental Panel on Climate Change.
- Huss, M., & Hock, R. (2015). A new model for global glacier change and sea-level rise. Frontiers of Earth Science, 3. 54. https://doi. org/10.3389/feart.2015.00054
- Kääb, A., Berthier, E., Nuth, C., Gardelle, J., & Arnaud, Y. (2012). Contrasting patterns of early twenty-first-century glacier mass change in the Himalayas. *Nature*, 488(7412), 495–498. https://doi.org/10.1038/nature11324
- King, O., Bhattacharya, A., Bhambri, R., & Bolch, T. (2019). Glacial lakes exacerbate Himalayan glacier mass loss. Scientific Reports, 9(1), 18145. https://doi.org/10.1038/s41598-019-53733-x
- Kraaijenbrink, P. D. A., Bierkens, M. F. P., Lutz, A. F., & Immerzeel, W. W. (2017). Impact of a global temperature rise of 1.5 degrees Celsius on Asia's glaciers. *Nature*, 549(7671), 257–260. https://doi.org/10.1038/nature23878
- Malz, P., Seehaus, T., Farías-Barahona, D., & Braun, M. (2019). Global glacier surface elevation change and geodetic mass balance estimations derived by spaceborne digital elevation models. Paper presented at the EGU General Assembly 2019. Austria.
- Maussion, F., Butenko, A., Champollion, N., Dusch, M., Eis, J., & Fourteau, K. (2019). The Open Global Glacier Model (OGGM) v1.1. Geoscientific Model Development, 12, 909–931. https://doi.org/10.5194/gmd-12-909-2019
- Mihalcea, C., Brock, B. W., Diolaiuti, G., D'Agata, C., Citterio, M., Kirkbride, M. P., et al. (2008). Using ASTER satellite and ground-based surface temperature measurements to derive supraglacial debris cover and thickness patterns on Miage Glacier (Mont Blanc Massif, Italy). Cold Regions Science and Technology, 52(3), 341–354. https://doi.org/10.1016/j.coldregions.2007.03.004
- Miles, E. S., Willis, I., Buri, P., Steiner, J. F., Arnold, N. S., & Pellicciotti, F. (2018). Surface pond energy absorption across four Himalayan glaciers accounts for 1/8 of total catchment ice loss. *Geophysical Research Letters*, 45(19), 10464–10473. https://doi. org/10.1029/2018GL079678
- Millan, R., Mouginot, J., Rabatel, A., Jeong, S., Cusicanqui, D., Derkacheva, A., & Chekki, M. (2019). Mapping surface flow velocity of glaciers at regional scale using a multiple sensors approach. *Remote Sensing*, *11*(21), 2498. https://doi.org/10.3390/rs11212498
- Nicholson, L., & Benn, D. I. (2012). Properties of natural supraglacial debris in relation to modeling sub-debris ice ablation. *Earth Surface Processes and Landforms*, 38(5), 490–501. https://doi.org/10.1002/esp.3299
- Nicholson, L., & Boxall, K. (2020). Supraglacial debris thickness measurements from excavation pits at Suldenferner. Zenodo. https://doi. org/10.5281/zenodo.3711581
- Østrem, G. (1959). Ice melting under a thin layer of moraine, and the existence of ice cores in moraine ridges. *Geografiska Annaler*, 41(4), 228–230. https://doi.org/10.1080/20014422.1959.11907953

Pellicciotti, F., Stephan, C., Miles, E., Herreid, S., Immerzeel, W. W., & Bolch, T. (2015). Mass-balance changes of the debris-covered gla-

ciers in the Langtang Himal, Nepal, from 1974 to 1999. Journal of Glaciology, 61(226), 373–386. https://doi.org/10.3189/2015JoG13J237 RGI Consortium (2017). Randolph glacier inventory—A dataset of global glacier outlines: Version 6.0: Technical Report. Boulder, CO: Global Land Ice Measurements from Space. https://doi.org/10.7265/N5-RGI-60

- Rounce, D. R., Byers, A. C., Byers, E. A., & McKinney, D. C. (2017). Brief communication: Observations of a glacier outburst flood from Lhotse Glacier, Everest area, Nepal. *The Cryosphere*, 11(1), 443–449. https://doi.org/10.5194/tc-11-443-2017
- Rounce, D. R., Khurana, T., Short, M. B., Hock, R., Shean, D. E., & Brinkerhoff, D. J. (2020). Quantifying parameter uncertainty in a largescale glacier evolution model using Bayesian inference: Application to High Mountain Asia. *Journal of Glaciology*, 66(256), 175–187. https://doi.org/10.1017/jog.2019.91
- Rounce, D. R., King, O., McCarthy, M., Shean, D. E., & Salerno, F. (2018). Quantifying debris thickness of debris-covered glaciers in the Everest region of Nepal through inversion of a subdebris melt model. *Journal of Geophysical Research: Earth Surface*, 123(5), 1094–1115. https://doi.org/10.1029/2017jf004395
- Rounce, D. R., & McKinney, D. C. (2014). Debris thickness of glaciers in the Everest area (Nepal Himalaya) derived from satellite imagery using a nonlinear energy balance model. *The Cryosphere*, 8(4), 1317–1329. https://doi.org/10.5194/tc-8-1317-2014
- Rounce, D. R., Quincey, D. J., & McKinney, D. C. (2015). Debris-covered glacier energy balance model for Imja-Lhotse Shar Glacier in the Everest region of Nepal. *The Cryosphere*, 9(6), 2295–2310. https://doi.org/10.5194/tc-9-2295-2015

Rowan, A., & Gibson, M. (2020). Supraglacial debris thickness data from Khumbu Glacier. Zenodo. https://doi.org/10.5281/zenodo.3775571
Scherler, D., Wulf, H., & Gorelick, N. (2018). Global assessment of supraglacial debris-cover extents. Geophysical Research Letters, 45(11), 11798. https://doi.org/10.1029/2018gl080158

- Seehaus, T., Malz, P., Sommer, C., Lippl, S., Cochachin, A., & Braun, M. (2019). Changes of the tropical glaciers throughout Peru between 2000 and 2016—Mass balance and area fluctuations. *The Cryosphere*, 13(10), 2537–2556. https://doi.org/10.5194/tc-13-2537-2019
- Shean, D. E., Bhushan, S., Montesano, P., Rounce, D. R., Arendt, A., & Osmanoglu, B. (2020). A systematic, regional assessment of high mountain Asia glacier mass balance. Frontiers of Earth Science, 7, 363. https://doi.org/10.3389/feart.2019.00363
- Shugar, D. H., Burr, A., Haritashya, U. K., Kargel, J. S., Watson, C. S., Kennedy, M. C., et al. (2020). Rapid worldwide growth of glacial lakes since 1990. Nature Climate Change, 10, 939. https://doi.org/10.1038/s41558-020-0855-4
- Sommer, C., Malz, P., Seehaus, T., Lippl, S., Zemp, M., & Braun, M. (2020). Rapid glacier retreat and downwasting throughout the European Alps in the early 21st century. *Nature Communications*, 11, 3209. https://doi.org/10.1038/s41467-020-16818-0
- Wu, Z., & Liu, S. (2012). Imaging the debris internal structure and estimating the effect of debris layer on ablation of glacier ice. Journal of the Geological Society of India, 80, 825–835.

References From the Supporting Information

Anderson, L., Wetterauer, K., Gök, D., Wahl, J., Dennis, D., & Scherler, D. (2020). Debris thickness measurements from various glaciers in the Western Alps. Zenodo. https://doi.org/10.5281/zenodo/4317470

- Anderson, L. S., Armstrong, W. H., Anderson, R. S., & Buri, P. (2021). Debris cover and the thinning of Kennicott Glacier, Alaska: In situ measurements, automated ice cliff delineation and distributed melt estimates. *The Cryosphere*, 15(1), 265–282. https://doi.org/10.5194/ tc-15-265-2021
- Banerjee, A., & Wani, B. A. (2018). Exponentially decreasing erosion rates protect the high-elevation crests of the Himalaya. Earth and Planetary Science Letters, 497, 22–28. https://doi.org/10.1016/j.epsl.2018.06.001
- Bocchiola, D., Senese, A., Mihalcea, C., Mosconi, B., D'Agata, C., Smiraglia, C., & Diolaiuti, G. (2015). An ablation model for debris-covered ice: The case study of Venerocolo Glacier (Italian Alps). *Geografia Fisica E Dinamicca Quaternaria*, 38, 113–128. https://doi.org/10.4461/GFDQ.2015.38.11



- Brock, B. W., Mihalcea, C., Kirkbride, M. P., Diolaiuti, G., Cutler, M. E. J., & Smiraglia, C. (2010). Meteorology and surface energy fluxes in the 2005–2007 ablation seasons at the Miage debris-covered glacier, Mont Blanc Massif, Italian Alps. *Journal of Geophysical Research*, 115(D9). https://doi.org/10.1029/2009jd013224
- Brock, B. W., Willis, I. C., & Sharp, M. J. (2006). Measurement and parameterization of aerodynamic roughness length variations at Haut Glacier d'Arolla, Switzerland. *Journal of Glaciology*, 52(177), 281–297. https://doi.org/10.3189/172756506781828746
- Brook, M. S., Hagg, W., & Winkler, S. (2013). Debris cover and surface melt at a temperate maritime alpine glacier: Franz Josef Glacier, New Zealand. New Zealand Journal of Geology and Geophysics, 56(1), 27–38. https://doi.org/10.1080/00288306.2012.736391
- Brook, M. S., & Paine, S. (2012). Ablation of ice-cored moraine in a humid, maritime climate: Fox Glacier, New Zealand. *Geografiska Annaler*, 94A(3), 339–349. https://doi.org/10.1111/j.1468-0459.2011.00442.x
- Brun, F., Buri, P., Miles, E. S., Wagnon, P., Steiner, J., Berthier, E., et al. (2016). Quantifying volume loss from ice cliffs on debris-covered glaciers using high-resolution terrestrial and aerial photogrammetry. *Journal of Glaciology*, 62(234), 684–695. https://doi.org/10.1017/ jog.2016.54
- Buri, P., Pellicciotti, F., Steiner, J. F., Miles, E. S., & Immerzeel, W. W. (2016). A grid-based model of backwasting of supraglacial ice cliffs on debris-covered glaciers. Annals of Glaciology, 57(71), 199–211. https://doi.org/10.3189/2016AoG71A059
- C3S (Copernicus Climate Change Service). (2017). ERA5: Fifth generation of ECMWF atmospheric reanalyses of the global climate. Copernicus Climate Change Service Climate Data Store. (CDS).
- Chand, M. B., Kayastha, R. B., Parajuli, A., & Mool, P. K. (2015). Seasonal variation of ice melting on varying layers of debris of Lirung Glacier, Langtang Valley, Nepal. Proceedings of the International Association of Hydrological Sciences, 368, 21–26. https://doi.org/10.5194/ piahs-368-21-2015
- Fyffe, C. L., Woodget, A. S., Kirkbride, M. P., Deline, P., Westoby, M. J., & Brock, B. W. (2020). Processes at the margins of supraglacial debris cover: Quantifying dirty ice ablation and debris redistribution. *Earth Surface Processes and Landforms*. https://doi.org/10.1002/ esp.4879
- Girod, L., Nuth, C., Kääb, A., McNabb, R., & Galland, O. (2017). MMASTER: Improved ASTER DEMs for Elevation Change Monitoring. *Remote Sensing*, 9(7). https://doi.org/10.3390/rs9070704
- Groos, A. R., Mayer, C., Smiraglia, C., Diolaiuti, G., & Lambrecht, A. (2017). A first attempt to model region-wide glacier surface mass balances in the Karakoram: Findings and future challenges. *Geografia Fisica E Dinamicca Quaternaria*, 40, 137–159. https://doi.org/10.4461/GFDQ.2017.40.10
- Hagg, W., Mayer, C., Lambrecht, A., & Helm, A. (2008). Sub-debris melt rates on southern Inylchek glacier, central Tian Shan. *Geografiska* Annaler: Series A, Physical Geography, 90(1), 55–63. https://doi.org/10.1111/j.1468-0459.2008.00333.x
- Han, H., Wang, J., Junfeng, W., & Liu, S. (2010). Backwasting rate on debris-covered Koxkar glacier, Tuomuer mountain, China. Journal of Glaciology, 56(196), 287–296. https://doi.org/10.3189/002214310791968430
- Han, H., Yongjian, D., & Shiyin, L. (2006). A simple model to estimate ice ablation under a thick debris layer. *Journal of Glaciology*, 52(179), 528–536. https://doi.org/10.3189/172756506781828395
- Hock, R (2005). Glacier melt: A review of processes and their modelling. Progress in Physical Geography: Earth and Environment, 29(3), 362–391. https://doi.org/10.1191/0309133305pp453ra
- Juen, M., Mayer, C., Lambrecht, A., Han, H., & Liu, S. (2014). Impact of varying debris cover thickness on ablation: A case study for Koxkar Glacier in the Tien Shan. *The Cryosphere*, 8(2), 377–386. https://doi.org/10.5194/tc-8-377-2014
- Juen, M., Mayer, C., Lambrecht, A., Wirbel, A., & Kueppers, U. (2013). Thermal properties of a supraglacial debris layer with respect to lithology and grain size. *Geografiska Annaler: Series A, Physical Geography*, 95(3), 197–209. https://doi.org/10.1111/geoa.12011
- Kayashta, R. B., Takeuchi, Y., Nakawo, M., & Ageta, Y. (2000). Practical prediction of ice melting beneath various thickness of debris cover on Khumbu Glacier, Nepal, using a positive degree-day factor. Debris-covered glaciers (Vol. 264, pp. 71–82). IAHS Publications.
- Klok, E. J., & Oerlemans, J. (2002). Model study of the spatial distribution of the energy and mass balance of Morteratschgletscher, Switzerland. Journal of Glaciology, 48(163), 505–518. https://doi.org/10.3189/172756502781831133
- Korona, J., Berthier, E., Bernard, M., Rémy, F., & Thouvenot, E. (2009). SPIRIT. SPOT 5 stereoscopic survey of Polar Ice: Reference Images and Topographies during the fourth International Polar Year (2007–2009). *ISPRS Journal of Photogrammetry and Remote Sensing*, 64(2), 204–212. https://doi.org/10.1016/j.isprsjprs.2008.10.005
- Lambrecht, A., Mayer, C., Hagg, W., Popovnin, V., Rezepkin, A., Lomidze, N., & Svanadze, D. (2011). A comparison of glacier melt on debris-covered glaciers in the northern and southern Caucasus. *The Cryosphere*, 5(3), 525–538. https://doi.org/10.5194/tc-5-525-2011
- Lejeune, Y., Bertrand, J.-M., Wagnon, P., & Morin, S. (2013). A physically based model of the year-round surface energy and mass balance of debris-covered glaciers. Journal of Glaciology, 59(214), 327–344. https://doi.org/10.3189/2013JoG12J149
- McCarthy, M., Pritchard, H., Willis, I. A. N., & King, E. (2017). Ground-penetrating radar measurements of debris thickness on Lirung Glacier, Nepal. Journal of Glaciology, 63(239), 543–555. https://doi.org/10.1017/jog.2017.18
- Mihalcea, C., Mayer, C., Diolaiuti, G., Lambrecht, A., Smiraglia, C., & Tartari, G. (2006). Ice ablation and meteorological conditions on the debris-covered area of Baltoro glacier, Karakoram, Pakistan. *Annals of Glaciology*, *43*, 292–300. https://doi. org/10.3189/172756406781812104
- Minora, U., Senese, A., Bocchiola, D., Soncini, A., D'agata, C., Ambrosini, R., et al. (2015). A simple model to evaluate ice melt over the ablation area of glaciers in the Central Karakoram National Park, Pakistan. Annals of Glaciology, 56(70), 202–216. https://doi. org/10.3189/2015AoG70A206
- Moore, P. L., Nelson, L. I., & Groth, T. M. D. (2019). Debris properties and mass-balance impacts on adjacent debris-covered glaciers, Mount Rainier, USA. Arctic, Antarctic, and Alpine Research, 51(1), 70–83. https://doi.org/10.1080/15230430.2019.1582269
- Nicholson, L., & Benn, D. I. (2006). Calculating ice melt beneath a debris layer using meteorological data. *Journal of Glaciology*, 52(178), 463–470. https://doi.org/10.3189/172756506781828584
- Nicholson, L. I., McCarthy, M., Pritchard, H. D., & Willis, I. (2018). Supraglacial debris thickness variability: Impact on ablation and relation to terrain properties. *The Cryosphere*, 12(12), 3719–3734. https://doi.org/10.5194/tc-12-3719-2018
- Nuth, C., & Kääb, A. (2011). Co-registration and bias corrections of satellite elevation data sets for quantifying glacier thickness change. *The Cryosphere*, *5*(1), 271–290. https://doi.org/10.5194/tc-5-271-2011
- Olson, M., Rupper, S., & Shean, D. E. (2019). Terrain induced biases in clear-sky shortwave radiation due to digital elevation model resolution for glaciers in complex terrain. *Frontiers in Earth Science*, 7. https://doi.org/10.3389/feart.2019.00216
- Patel, L. K., Sharma, P., Thamban, M., Singh, A., & Ravindra, R. (2016). Debris control on glacier thinning—a case study of the Batal glacier, Chandra basin, Western Himalaya. *Arabian Journal of Geosciences*, 9(4). https://doi.org/10.1007/s12517-016-2362-5
- Porter, C., Morin, P., Howat, I. M., Noh, M.-J., Bates, B., Peterman, K., et al. (2018). ArcticDEM. Harvard Dataverse, V1. https://doi.org/10.7910/DVN/OHHUKH



- Quincey, D., Smith, M., Rounce, D., Ross, A., King, O., & Watson, C. (2017). Evaluating morphological estimates of the aerodynamic roughness of debris covered glacier ice. *Earth Surface Processes and Landforms*, 42(15), 2541–2553. https://doi.org/10.1002/esp.4198
 Reid, T. D., & Brock, B. W. (2010). An energy-balance model for debris-covered glaciers including heat conduction through the debris layer.
- Journal of Glaciology, 56(199), 903–916. https://doi.org/10.3189/002214310794457218
 - Reid, T. D., & Brock, B. W. (2014). Assessing ice-cliff backwasting and its contribution to total ablation of debris-covered Miage glacier, Mont Blanc massif, Italy. *Journal of Glaciology*, 60(219), 3–13. https://doi.org/10.3189/2014JoG13J045
 - Reid, T. D., Carenzo, M., Pellicciotti, F., & Brock, B. W. (2012). Including debris cover effects in a distributed model of glacier ablation. Journal of Geophysical Research: Atmospheres, 117(D18). https://doi.org/10.1029/2012jd017795
 - Shah, S. S., Banerjee, A., Nainwal, H. C., & Shankar, R. (2019). Estimating of the total sub-debris ablation from point-scale ablation data on a debris-covered glacier. *Journal of Glaciology*, 65(253), 759–769. https://doi.org/10.1017/jog.2019.48
 - Sommer, C., Malz, P., Seehaus, T., Lippl, S., Zemp, M., & Braun, M. (2020). Rapid glacier retreat and downwasting throughout the European Alps in the early 21st century. *Nat Commun.* https://doi.org/10.1038/s41467-020-16818-0
 - Tarboton, D. G., & Luce, C. H. (1996). Utah energy balance snow accumulation and melt model (UEB). Intermountain Research Station: Utah Water Research Laboratory, Utah University and USDA Forest Service.
 - Wang, L., Li, Z., & Wang, F. (2011). Spatial distribution of the debris layer on glaciers of the Tuomuer Peak, western Tian Shan. Journal of Earth Science, 22(4), 528–538. https://doi.org/10.1007/s12583-011-0205-6
 - Wang, P., Li, Z., Li, H., Wang, W., Zhou, P., & Wang, L. (2017). Characteristics of a partially debris-covered glacier and its response to atmospheric warming in Mt. Tomor, Tien Shan, China. *Global and Planetary Change*, 159, 11–24. https://doi.org/10.1016/j. gloplacha.2017.10.006
 - Watson, C. S., Quincey, D. J., Smith, M. W., Carrivick, J. L., Rowan, A. V., & James, M. R. (2017). Quantifying ice cliff evolution with multi-temporal point clouds on the debris-covered Khumbu Glacier, Nepal. Journal of Glaciology, 63(241), 823–837. https://doi. org/10.1017/jog.2017.47
 - Yang, W., Yao, T., Baiqing, X., & Zhou, H. (2010). Influence of supraglacial debris on summer ablation and mass balance in the 24k Glacier, Southeast Tibetan Plateau. *Geografiska Annaler*, 92A(3), 353–360. https://doi.org/10.1111/j.1468-0459.2010.00400.x
 - Yang, W., Yao, T., Zhu, M., & Wang, Y. (2017). Comparison of the meteorology and surface energy fluxes of debris-free and debris-covered glaciers in the southeastern Tibetan Plateau. *Journal of Glaciology*, *63*(242), 1090–1104. https://doi.org/10.1017/jog.2017.77
 - Zhang, Y., Fujita, K., Liu, S., Liu, Q., & Nuimura, T. (2011). Distribution of debris thickness and its effect on ice melt at Hailuogou glacier, southeastern Tibetan Plateau, using in situ surveys and ASTER imagery. *Journal of Glaciology*, 57(206), 1147–1157. https://doi. org/10.3189/002214311798843331